Author's response to reviews

Title: The role of paternal psychosocial work condition on mental health of their children: A case-control study

Authors:

Stefania Maggi (stefania.maggi@gmail.com)
Aleck Ostry (ostry@interchange.ubc.ca)
James Tansey (james.tansey@sbs.ox.ac.uk)
James Dunn (jim.dunn@utoronto.ca)
Ruth Hershler (hershler@interchange.ubc.ca)
Lisa Chen (lisachen@interchange.ubc.ca)
Clyde Hertzman (hertzman@interchange.ubc.ca)

Version: 4 Date: 7 March 2008

Author's response to reviews: see over
BMC Editorial Team

March 6, 2008

Dear Editor,

We are pleased to send you the accepted manuscript #6691101541290657, to which we have made minor revisions as suggested by reviewer 2.

Specifically, on pg. 14 (in the Discussion section) we have added to our discussion on the role ethnicity in developmental psychopathology.

We look forward to hear from you on details of the publication process for this manuscript. We also take this opportunity to thank the two reviewers for their insightful comments that have contributed significantly to the quality of this manuscript.

Sincerely,

Stefania Maggi.